
 “CL31 Replacement with ASOS ACU V2.79V and DCP V2.0 EPROM” OT&E  
Test Review Group Status Meeting  
 
March 19, 2009 
 
ATTENDEES: 
 
NWS: 
OT&E Site Electronics System Analyst (ESA)/ Electronic Technician (ET): 
WFO Blacksburg, VA 
WFO Midland, TX 
WFO Dodge City, KS 
WFO Cheyenne, WY 
WFO Hilo, HI 
 
NWS Regional Headquarters: 
Ed Doerr, ARH  
John Bush PRH 
Lew Harrington, SRH RMS 
Dan Lester, CRH  RMS 
Matt Ferrell , ERH  RMS (ABSENT) 
Kevin Murray, ERH (ABSENT) 
Tim Rutkowski, ERH (ABSENT) 
Son Nguyen, WRH (ABSENT) 
 
National Weather Service Headquarters: 
John Monte – W/OST11 
Joel Williams – W/OST11 (ABSENT) 
Joseph Devost – W/OPS12 
David Mannarano – W/OPS22  
Richard Parry - W/OPS22 
Chet Schmitt – W/OPS22 
Peggy Hoch – W/OPS23 
Hak Kim – W/OPS23  
Kevin Conaty – AMOC (ABSENT) 
Beth McNutly – W/OS23 
Laura Cook – W/OS7 (ABSENT) 
Jennifer Dover – W/OPS22 
Brian Rice – SAIC Contractor – Sterling Field Support Center, Sterling, VA 
Khien Nguyen – W/OPS24 
Joseph Fiore – W/OPS24  
Harry Tran – W/OPS24 
Jerald Dinges – W/OPS24 (Moderator) 
 
National Weather Service Training Center 
Bob Retzlaff  - Kansas City, MO 
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NWS Employees Organization (NWSEO) Representative: 
Chris Kornkven –WFO Milwaukee, WI  
 
FAA: 
Bing Huang, ATO- T  
Jerry Kranz, (contractor) 
 
DoD - U.S. Navy: 
Ronald Heatherdale and Gerald “Wayne” Knight – Space and Naval Warfare System 
Center (SPAWARSYSCEN), Charleston, SC  (ABSENT) 
 
DoD - U.S. Air Force: 
William (Mac) Lawrence 
 
The second CL31 ceilometer replacement Operational Test and Evaluation (OT&E) Test 
Review Group (TRG) status meeting was held after the 03/09/09 OT&E Readiness 
Review meeting by audio conference call and “GO TO MEETING” web-based software 
on Thursday, March 19, 2009.  Joseph Fiore, OT&E Director, convened the meeting with 
a “roll call” (See above list of attendees). 
 
Joe presented the agenda for the meeting as follows (The noteworthy discussions and 
statements are included): 
 

1) Ratify Thursday, March 12th minutes.  – These were ratified by the TRG. The 
minutes will be posted to the W/OPS24 webpage. 

2) Review of the “action items” from the previous minutes - Those still in the 
“OPEN” status will remain unchanged until some milestone occurs in the OT&E. 
These will be reviewed again when the associated OT&E milestone occurs.   

 
The following is the accounting for each action item: 
 

a. Action Item 2 (11/17/08) remains OPEN.  
b. Action Item 3 (11/17/08) remains OPEN (No change until Phase II OT&E). 
c. Action Item 6 (11/17/08) remains (No change until all sites report successful 

installations). 
d. Action Item 7 (11/17/08) remains OPEN. (No change until all OT&E sites 

successfully install the CL31 and the ASOS ACU V2.79V and V2.0 DCP 
EPROMS) 

e. Action Item 8 (03/09/09) remains OPEN.  (No change until OT&E is 
completed). 

f. Action Item 10 (03/12/09) CLOSED.  
g. Action Item 11 (03/12/09) remains OPEN. (No change until all installations 

for the 22 OT&E sites is completed). 
 

3) Review of the ASOS CL31 ceilometer and ASOS firmware/EPROMs installation 
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at the 22 OT&E sites -  Joe Fiore reported the following sites successfully 
installed ASOS V2.79V including the CL31 (+ CL31) - NOTE: Those noted (*) 
without a “+ CL31” previously installed the CL31 as part of the CL31/CT12K 
Meteorological Comparison Evaluation: 
 

• 03/12/09 18:05:45   BIS - ASOS version 2.79V (*)  
• 03/13/09 13:18:53  KNBC - ASOS version 2.79V  + CL31 
• 03/16/09 18:04:08   ABR - ASOS version 2.79V  + CL31 
• 03/16/09 19:23:41   CAR - ASOS version 2.79V  + CL31 
• 03/16/09 21:56:57   ITO - ASOS version 2.79V (*) 
• 03/18/09 15:50:35   ANJ - ASOS version 2.79V  + CL31 
• 03/18/09 16:19:30   GDP - ASOS version 2.79V (*) 
• 03/18/09 19:01:53   GUY - ASOS version 2.79V  + CL31 
• 03/19/09 16:20:02   CYS - ASOS version 2.79V  + CL31 
• 03/19/09 18:19:10   HIO - ASOS version 2.79V (*)(Note: reported 

after the meeting) 

NOTE:  Joe will provide an updated table of the status of all CL31 and V2.79V 
installations as an attachment to these minutes and subsequent minutes during the 
OT&E installations. [See below Action Item 11 (03/12/09)] 
 
Joe also reminded sites to provide the serial numbers of the CL31s installed.  
However, WFO Blacksburg, VA focal point noted that when the CL31 is installed, 
the Electronic Technician is required to include the serial number when  they 
document the installation completion in the Engineering Maintenance Reporting 
System (EMRS).  John Monte will access EMRS to obtain the serial numbers and 
will e-mail the site technician  for the information only if he can’t find it in EMRS. 
 
4) Report from the NWS regional headquarters and site focal points on problems 

encountered –  
a. Bob Retzlaff reported he installed V2.79V on his SCA system this week.  

He has not seen any error messages since the installation.  However, BIS 
which was error-free since installing V2.79V Thursday, March 12, had 
two AWPAG  sensor time out SYSLOG error messages recently.  This 
lead to the report of problems seen also at the two other SCA OT&E sites 
GUY and ANJ (See below discussions).   

b. Dan Lester, CRH, reported that Ste. St. Marie, MI ASOS SCA had a 
sensor timeout for the CT12K (SYSLOG error #1015) after V2.79V was 
installed on Wednesday, March 18.  They had moved the CT12K to its 
own SIO (Port 2 with Port 1 empty).  They also at Joe DeVost’s 
recommendation ordered a new CT12K from NLSC and will replace the 
existing CT12K and see if resolves the 1015 error message.  It should be 
noted they had sensor time out problems with the CT12K interface to the 
SCA prior to the installation of ASOS ACU V2.79V.   

c. Bill Cronin, WFO Cheyenne, WY (CYS), reported he could not get power 
to the CL31 and he received damaged V2.0 DCP EPROMs.  John Monte 
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stated the CL31 problem should be considered “infant mortality” since it 
is a brand new device. Therefore, he requested the CL31 be repackaged 
and sent to John.  John will coordinate with SFSC and Jerry Dinges on 
what replacement CL31 could be sent to WFO CYS.  After the meeting 
John and Jerry agreed to send one of the CL31 from the ASOS test system 
ST0 at Sterling Field Support Center, Sterling, VA.  A pre-production 
CL31 will replace the CL31 removed from ST0.  John Monte will 
coordinate this activity.  Also, Joe Devost, W/OPS12, will send WFO 
CYS a new set of V2.0 DCP EPROMS. 

d. Lew Harrrington, SRH, reported the GUY is having problems with the 
CT12K communications since V2.79V was installed. Bob Retzlaff, 
NWSTC, called Lew after the meeting and recommended the site reinstall 
V2.79V.  Bob had the same problem with his SCA with V2.79D.  The 
problem went away when he reinstalled V2.79D. 

e. A question arose from the CYS problem whether if problems with the 
CL31 what is the sparing philosophy for the CL31.  John Monte stated that 
there is only 1 spare, the CL31 processor board.  The NWS Electronic 
Technician will be provided 1 measuring unit since that is the only 
hardware component failing. 

f. Another question presented, what was the disposal procedure for the 
CT12K?  Rick Parry, W/OPS22, stated, according to the CL31 
Deployment Plan, at the completion of the OT&E, the CT12K should be 
returned to NLSC for reconditioning and put into spares. Once enough 
spares are received, the sites will be asked to dispose of the CT12K 
locally.  A concern was raised “local site disposal” puts the cost of the 
disposal on the NWS regions.  An action was requested to have the 
Maintenance Branch (W/OPS12) and the Logistics Branch (OPS12) to 
provide a disposal plan for the CT12K.   

 
5) John Monte requested the TRG /ATRB members opinion if there were any risks 

or problems with the CL31 -   John stated, as CL31 Project Manager, he will be 
participating in a Production Readiness Review meeting Friday, March 20 at 1 
p.m. EDT with Vaisala representatives for the ASOS CL31 replacement 
ceilometer.   He requested comments be e-mailed to him. 

6) Joe Devost, W/OPS12, reported he completed an action to send a spreadsheet 
containing the hardware deliverables needed to install the CL31.   A discussion 
followed about ASOS needing additional SIO cards and/or “A/D cables if there 
are more than 9 sensors interfaced to the ASOS.  The NWS regional headquarters 
wanted Joe to survey all ASOSs for this requirement.  However, Joe noted that 
this is only a problem with the OT&E sites because both the CT12K and CL31 
ceilometers need to be interfaced to the ASOS during Phase I of the OT&E.  Joe 
believes he has a good “handle” on the OT&E sites requirements. 

 
The final agenda topic for the TRG was the status/ “wrap-up of the System Test (ST) for 
the ASOS Ice-free Wind (IFW) sensor firmware change V4.54. 
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7) Wrap-up for the IFW sensor V4.54 ST -  Harry Tran (W/OPS24) presented a 
synopsis of the 2-week test which began Tuesday, March 10 at SFSC, Sterling, 
VA and concluded Wednesday, March 18.   The purpose of the ST was to verify 
the draft NWS Modification Note for the installation of the sensor firmware and 
to verify the fix works in both the current ASOS ACU operational firmware 
V2.79D and the proposed operational firmware V2.79V using both a good IFW 
sensor and one with a faulty thermistor.  The fix corrects a problem where the 
ASOS maintenance screen was not displaying a “failed” flag for the IFW sensor 
and no System Log error message was generated.  All tests configurations and 
regression tests were successful.  On Test Trouble Report was generated, #210 
against the ASOS V2.79V firmware.  Rather than the ASOS maintenance page 
displaying an “F” for a failed IFW sensor, it displays an “*.”  The correct “F” 
displays with V2.79D the current operational ASOS firmware.  There is no $ 
generated with the V2.79D and the wind data is missing.  With the wind data 
missing, the $ should be generated.  Without a $, the Electronics Technician will 
not know of the need to fix the problem.  The TRG stated it is unacceptable to 
deploy the IFW V4.54 sensor firmware with V2.79V as an operational ASOS 
firmware.  Peggy Hoch, W/OPS23 will get a proposal from PRISM, Inc, ASOS 
software contractor.  Richard Parry, W/OPS22, was tasked to identify possible 
beta test sites to validate the sensor firmware does not degrade the ASOS 
performance prior to national deployment.  Once the PRISM proposal is received, 
the ASOS Configuration Control Board will determine how the ASOS ACU fix 
will be implemented and to allow IFW V4.54 to be deployed. 

   
The following is a status on the OPEN Action Items assigned: 
 
Action Item 2 (11/17/08) - OPEN:  Assigned to W/OPS12 regarding any SIO/ATD 
boards sent to Electronic Technicians. W/OPS12 will provide W/OPS24 with a list of 
OT&E sites that require an extra “ATD” board if they have more than eight sensors on 
one ATD board. This list and explanation for the extra ATD board will be e-mailed to the 
site requiring the extra board by W/OPS24. 
 
Action Item 3 (11/17/09) - OPEN:  Assigned to W/OPS12 and W/OPS24 regarding 
when Phase II OT&E sites will get their additional CL31. An e-mail will be sent to ROA, 
CMH, OKC, and PHX to inform them when to except the second ceilometers. ROA, 
CMH, OKC, and PHX will receive their second CL31 ceilometer after the decision is 
made to make the CL31 the operational ceilometer. 
 
STATUS:   Report on this action item until just prior to PHASE II OT&E 
commencement    
 
Action Item 6 (11/17?09) – OPEN:  Assigned to ESA’s to inventory all items they 
receive from NLSC and WSH. The ESA’s will inventory all items received (CL31 
ceilometer) CL31 hardware kits, V2.0 DCP EPROMS, V2.79V software (and documents 
obtained from the OPS12 website), and let the OT&E Test Director (Joe Fiore) know by 
e-mail (or phone) if the inventory list is complete.   
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STATUS:  This action will remain OPEN until all 22 OT&E sites have the CL31 
ceilometer installed along with the ASOS ACU V2.79V firmware and DCP V2.0 
EPROMS. 
 
Action Item 7 (11/17/09) - OPEN:   Assigned to ESA’s. The ESA’s (NOTE:  exception 
is Dan Lester for NWS Central Region) will notify the OT&E Test Director (Joe Fiore) 
by e-mail when they are ready to install the CL31 ceilometer, V2.0 DCP EPROMS (for 
sites that use EPROMS), and ASOS V2.79V ACU Software. They will also notify the 
OT&E test director when installation of all required material is complete. 
 
STATUS:   This action will remain OPEN until all 22 OT&E sites have the CL31 
ceilometer installed along with the ASOS ACU V2.79V firmware and DCP V2.0 
EPROMS. 
 
Action Item 8 (03/09/09) - OPEN:   National Weather Service Headquarters will 
download the SYSLOG error messages (1015, 1515, and 1537) from all 22 OT&E sites 
daily and will analyze the data to determine how many random sensor time out SYSLOG 
error messages are generated with a $.  SAIC contract personnel at Sterling Field Support 
Center, Sterling VA will perform the work. Jennifer Dover (W/OPS22) will report the 
statistics to the OT&E TRG at the weekly meetings during the OT&E.  In preparation for 
this activity SAIC personnel at SFSC will also analyze the same SYSLOG error 
messages for each 15 OT&E site (22 minus the 7 Meteorological Comparison Evaluation 
Beta sites) for 30 days PRIOR to the OT&E.  This analysis will “baseline” the number of 
random sensor time out errors occurrences before V2.79 V is installed. 
 
STATUS:  This action will remain open until the completion of the OT&E. 
 
Action Item 10 (03/12/09) – CLOSED:   W/OPS12 and W/OPS24 assigned the action.  
Joe DeVost will provide the updated spreadsheet listing the hardware items necessary to 
install the CL31 ceilometer for each OT&E site.  These items will be delineated by which 
are to be provided from WSH and which should be requisitioned from NLSC.  Joe Fiore 
will disseminate the spreadsheet to the appropriate OT&E sites who have NOT install the 
CL31 yet. COMPLETED 
 
Action Item 11 (03/12/09) – OPEN:  Assigned to W/OPS24.  Joe Fiore will provide the 
TRG a status update at each status meeting on the CL31 ceilometer and ASOS ACU 
V2.79V and DCP V2.0 EPROM installation until all 22 sites have completed this 
activity. 
 
STATUS:  This action will remain OPEN until all 22 OT&E sites have completed their 
installations. 
 
The new action items as a result of March 19th meeting are as follows: 
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Action Item 12 (0319/09) – OPEN:   Assigned to W/OST11.  John Monte will query 
EMRS to obtain the serial numbers for the CL31 installed at the 22 OT&E sites.  He will 
e-mail the appropriate contact if the information is not found in EMRS. 
 
Action Item 13 (0319/09) – OPEN:  Assigned to the WFO CYS.  Bill Cronin will 
repackage and ship the CL31 to John Monte W/OST11. 
 
Action Item 14 (03/19/09) – OPEN:  Assigned to W/OST11.  John Monte will ensure 
Bill Cronin is shipped a replacement CL31.  John will also coordinate with SFSC to 
obtaining and replacing one of the CL31 on the ASOS test system ST0. 
 
Action Item 15 (03/19/09) – OPEN:  Assigned to W/OPS12.  Joe DeVost will send Bill 
Cronin new ASOS DCP V2.0 EROMS. 
 
Action Item 16 (03/19/09) – OPEN:  Assigned to SRH and CRH ASOS focal points.   
Lew Harrington (SRH) and Dan Lester (CRH) will report to the TRG the status on the 
GUY and ANJ SCA communication problems between the V2.79V and the CT12K until 
resolved.   
 
Action Item 17 (03/19/09) – OPEN:   Assigned to W/OPS12 and W/OPS14.  Joe Devost 
and Fred Hauschildt will provide the disposal plan for the CT12K. 
 
Action Item 18 (09/19/09) – OPEN:   Assigned to the TRG/ATRB/NWS field sites.  
Send any comments regarding known risks or problem related to the CL31 performance 
by e-mail NLT Friday, March 20.  John requested comments as input to the Production 
Readiness Review meeting with Vaisala on the CL31 at 1 pm EDT. 
 
Action Item 19 (09/19/09) – OPEN:  Assigned to W/OPS23.  Peggy Hoch will request a 
proposal from PRISM, Inc. to fix a problem in ASOS ACU V2.79V firmware related to 
TTR#210 as documented during the ST for IFW V4.54 sensor firmware. 
 
Action Item 20 (09/19/09) –CLOSED:   Assigned to W/OPS22.  Rick Parry will provide 
Joe Fiore with candidate list of operational ASOS sites for a beta test of IFW V4.54. 
COMPLETED 
 
STATUS:  The list was provided on Thursday, March 19 in and e-mail. 
 
Action Item 21 (09/19/09) – OPEN:  Assigned to ASOS CCB.  The ASOS ACCB will 
determine the course of action once Prism’s proposal to fix ASOS ACU V2.79V related 
to the IFW V4.54 sensor firmware. 
 
The next OT&E TRG meeting will be scheduled for Thursday, March 26 at 2 pm EDT 
to provide a status report on OT&E activities.  There will only be an audio conference 
call.  Please use the following information to dial into the meeting: 
 
Telephone:  1-866-685-1879 
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Password:  8259362# 
 
. 
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